Author's response to reviews

Title: The effect of mode of delivery on HIV-1 disease progression and mortality in a Kenyan cohort

Authors:

Jennifer A Unger (junger@uw.edu)
Barbra A Richardson (barbrar@uw.edu)
Phelgona Otieno (POtieno@kenri.org)
Carey Farquhar (cfarq@uw.edu)
Dalton Wamalwa (Dalton@africaonline.co.ke)
Grace C John-Stewart (gjohn@uw.edu)

Version: 2
Date: 3 December 2013

Author's response to reviews:

It was requested that we make the following revisions:

*1 Statistical analysis: Please state the significance level you have used.

We have made the following changes to the manuscript:

#1: In all study analyses, P#0.05 was considered significant. This statement has been added to the statistical analyses section of the paper.

#2: a P value from a Student’s t-test for continuous variables and Pearson’s chi-square tests for categorical variables (statistical significance, # = 0.05). This statement has been added to Table 1 as a footnote.

We appreciate your comments and look forward to a positive response from your reviewers.